Author’s response to reviews

Title: Acute toxicity, Brine shrimp cytotoxicity, anthelmintic and relaxant potentials of fruits of Rubus fruticosus Agg.

Authors:

Niaz Ali (niazpharmacist@yahoo.com)
Umer Aleem (umaraleem@yahoo.com)
Syed Wadood Ali Shah (wadudalishah@yahoo.com)
Ismail Shah (ismailshah09@yahoo.com)
Muhammad Junaid (juni_phdr@yahoo.com)
Ghayour Ahmed (pahrmacist@gmail.com)
Waqar Ali (waqarmohmadzai@yahoo.com)
Mehreen Ghias (mehreenghias@yahoo.com)

Version: 4 Date: 3 June 2013

Author’s response to reviews: see over
Title: Acute toxicity, Brine shrimp cytotoxicity, anthelmintic and relaxant potentials of fruits of *Rubus fruticosus Agg.*

Version: 3 Date: 3rd June, 2013

Replies to the editorial comments

1. The full name of ethical committee has been written in the manuscript which has been highlighted yellow.
2. The language has been rectified through experts.
   Hope you will now give prompt decision.

Regards!